Author's response to reviews

Title: The Value of Radiographic Findings for the Progression of Pandemic 2009 Influenza A/H1N1 Virus Infection

Authors:

Takanori Funaki (funaki-t@ncchd.go.jp)
Kensuke Shoji (shoji-k@ncchd.go.jp)
Nobuyuki Yotani (yotani-n@ncchd.go.jp)
Tomohiro Katsuta (jesus@pf6.so-net.ne.jp)
Osamu Miyazaki (miyazaki-o@ncchd.go.jp)
Shunsuke Nosaka (nosaka-s@ncchd.go.jp)
Hidekazu Masaki (masaki-h@ncchd.go.jp)
Akihiko Saitoh (asaitoh@med.niigata-u.ac.jp)

Version: 5 Date: 5 October 2013

Author's response to reviews: see over
Oct 5, 2013

Philippa Harris, PhD
Executive Editor
BMC-series Journals
Biomed Central
Floor 6, 236 Gray’s Inn Road
London, WC1X 8HL

RE: Manuscript reference number: 3688456847264303, MINOR

Dear Dr. Harris:

Our manuscript, “The Value of Radiographic Findings for the Progression of Pandemic 2009 Influenza A/H1N1 Virus Infection,” has been revised according to the suggestions of the Editor. An item-by-item explanation of our responses to each query appears below. Please note that we paraphrased each of the Reviewers’ questions and suggestions in our responses.

Editor:

*Please also ensure that your revised manuscript conforms to the journal style.

We have revised the manuscript according to the BMC journal style. We have changed the authors’ information and a reference list formatted to the BMC journal style.

*What does your data look like if you concentrate only on the 139 patients with known influenza?

We have added the analyses in sub-groups of 139 laboratory confirmed patients in Result section. These results were similar to the results of total subjects. (page 13, line 9-17)

We thank the editor and reviewers for their important constructive comments on our
manuscript and hope you will find the revised manuscript acceptable for publication in the
BMC Infectious Diseases.

Thank you very much for your consideration.

Sincerely,

Takanori Funaki, MD
Fellow
Division of Pediatric Infectious Diseases
National Center for Child Health and Development
E-mail: funaki-t@ncchd.go.jp

Akihiko Saitoh, MD, PhD, FAAP
Professor and Chairman
Department of Pediatrics
Niigata University Graduate School of Medical and Dental Sciences
E-mail: asaitoh@med.niigata-u.ac.jp